Author's response to reviews

Title: Severe maternal morbidity and near miss: occurrence and associated factors in Sergipe, Northeast Brazil

Authors:

Larissa P.L. Galvão (lari.galvao@uol.com.br)  
Fabiano Alvim-Pereira (falvim@gmail.com)  
Caio M.M. Mendonça (menesescio@hotmail.com)  
Filipe E.F. Menezes (filipe.furao@gmail.com)  
Kaique A.N. Góis (kaique_001@hotmail.com)  
Ruy F. Ribeiro Jr. (ruy_junin@hotmail.com)  
Ricardo Q. Gurgel (ricardoggurgel@gmail.com)

Version: 3 Date: 23 July 2013

Author's response to reviews: see over
Aracaju, July 22nd 2013.

MS: 1199914686974625

The prevalence of severe maternal morbidity and near miss and associated factors in Sergipe, Northeast Brazil


Dear Editors,

Thank you for considering our paper for publication. We are very grateful for the helpful suggestions put forward by the reviewers and we have responded to each comment in this document. We have also revised the entire manuscript and believe that it has been improved significantly. We hope that you will now consider it to be suitable for publication, but we would be happy to make further changes if you feel that they are necessary.

Best wishes,

Ricardo Q. Gurgel

Corresponding author